Author's response to reviews

Title: Burden of morbidity in a patient perspective - the case of sick-leave certified patients in primary care

Authors:

Lennart Carlsson (lennart.carlsson@ki.se)
Roland Morgell (roland.morgell@ki.se)
Lars-Erik Strender (lars-erik.strender@ki.se)
Britt Arrelöv (britt.arrelov@sll.se)
Gunnar H Nilsson (gunnar.nilsson@ki.se)

Version: 8  Date: 22 May 2009

Author's response to reviews:

Dear Editor,

The final revisions are now made, all in the main text. Two minor misspellings. Legends to figures 1 and 2 no longer in table format, and table in table format "simple 1".

Thanks for your patience!